Author's response to reviews

Title: Cost-effectiveness of six strategies for Helicobacter pylori diagnosis and management in uninvestigated dyspepsia assuming a high resource intensity practice pattern

Authors:

Kyland P Holmes (kyland.holmes@gmail.com)
John C Fang (John.Fang@hsc.utah.edu)
Brian R Jackson (brian.jackson@aruplab.com)

Version: 3 Date: 24 November 2010

Author's response to reviews:

I made the formatting changes requested in the last email to me on 22 Nov 10:
*Title page formatting fixed
*Abstract removed "Aims" header
*Added brief "Conclusions section at end of manuscript
*Formatted tables to meet requirements
*Reformatted the figure legend (put it in the body of the text, removed from the image)
*Removed Figure 2 since it's not referenced in the final version of the manuscript
*Reformatted author contributions and affiliations